Author's response to reviews

Title: Spatial distribution of Parkinson's disease mortality in Spain, 1989-1998, as a guide for focused aetiological research or health-care intervention.

Authors:

Jesús de Pedro-Cuesta (jpedro@isciii.es)
Gonzalo López-Abente (glabente@isciii.es)
Eduard Rodríguez-Farré (erfnmc@iibb.csic.es)

Version: 2 Date: 18 June 2009

Author's response to reviews:

This paper may deserve publication in BMC PH due to the interest to a wide band of readers suggested by the double application field of results, namely, environmental research and development of primary care and neurological diagnosis. In addition, due to the length and the required detailed and coloured reproduction of maps, it would further fit for online access at BMC.